Accelerator Status Report

Run Schedule: March 11, 2010 to July 22, 2010
March 31, 2010
Injector    Energy 56.62 MeV
· (3/27) Injector beam instability: chicane dipole MDK0R02 had a ground fault due to water dripping into the magnet assembly. Beam motion was seen horizontally starting from IPM0R04. 

· (3/25) LCW valve got stuck, which caused the Capture power to be unstable. The problem was fixed. 
· Reboot of iocin3 caused the laser spot to move, which caused beam instability. The problem was corrected. After reboot of an ioc make sure all the parameters are back.
· Do not just acknowledge alarms. The alarms may indicate problems.
· Capture power is not stale, which may cause problems at some pint.

.
Accelerator   Linac Energy 503.25 MeV
· There were about three shifts of downtime this weekend. MDK0R02 dipole had a ground fault. Rack SL19B09 had an electrical problem. VBV2L08A would not open due to the faulty indicator switch.

· (3/29) Injector dropped to power permit due to storm.

· (3/28) Arc9 box tripped off once. 
· (3/27) NL RF gang phase shifted by 15 degrees. Do not know why yet. 
· Ops found 1L13 HPA LCW leak during a rad survey.
· (3/26) 100uA CW beam was established to Hall A dump.
· The accelerator has been running well with very low trip rate.

· Low energy run makes RF more stable.

· RF was economized to save power.

· From the FOPT data the optics in the main machine is alright but had been adjusted a couple of times when experts were trying to match Hall A line. There is dispersion leaking from the injector into NL which may affect Hall A. We have not sent high current in the machine yet.  High currents will be a real test of the machine optics.
· How to run CW beam to the BSY dump while Hall C is in access?

Hall A---Pass 1, Prex from March 11 to May 24, 2010
· Experts are still working trying to get the desired optics in Hall A line. The hall wants to have very small beam (50 um x 50 um). Currently the Moller quads are off.

· (3/30) Hall started commissioning its Moller.    
· MAT1H01H air core corrector was replaced with MBC1H01H iron core corrector to increase the steering strength. MBC1H01V polarity was wrong and corrected.
· IPM1H01, IPM1H04A and IPM1H04B were quad centered and the SOFs were corrected.

· BLA was calibrated to 50uA for Hall A. Ops can deliver up to 100uA but the hall wants to take 50uA for now. FFB is working in energy mode.
· A new corrector will be installed on Hall A beam pipe above Compton third section.

· Hall A Moller procedure was revised. It is separated from the hall beam delivery procedure. Ops will be moving Hall A Moller target during Moller runs. The hall may use different beam structure during Moller runs. Hall A may commission its Moller this week. Please review the Moller procedure.
· Experts have been trying to do Hall A match but the results are not good enough yet.

Hall B ---Pass 3, g9-Frost from March 11 to July 22, 2010
· Hall B beam was fine until yesterday. This morning the hall reported that after yesterday’s beam studies its beam was different. We are investigating. 
· The hall reported that beam was a bit unstable, most likely caused by the injector spot move after reboot of iocin3. After the injector checkup Hall B beam was much better.
Hall C---QWeak from May 25 to July 22, 2010
· QWeak installation
